#### AGU100 ADVANCING EARTH AND SPACE SCIENCE

# **JGR** Oceans



# **REVIEW ARTICLE**

10.1029/2019JC015330

# **Special Section:**

Atlantic Meridional Overturning Circulation: Reviews of Observational and Modeling Advances

#### **Key Points:**

- Theoretical models have provided a dynamical framework for understanding recent AMOC observations and numerical model results
- We review recent advances in conceptual understanding of the processes setting the strength, structure, and variability of the AMOC
- Recent models go beyond the zonal average, include interaction between wind and buoyancy forcing, and acknowledge the importance of eddies

Correspondence to: H. L. Johnson, helen.johnson@earth.ox.ac.uk

#### Citation:

Johnson, H. L., Cessi, P., Marshall, D. P., Schloesser, F., & Spall, M. A. (2019). Recent contributions of theory to our understanding of the Atlantic Meridional Overturning Circulation. *Journal of Geophysical Research: Oceans*, 124, 5376–5399. https://doi. org/10.1029/2019JC015330

Received 29 MAY 2019 Accepted 29 JUL 2019 Accepted article online 6 AUG 2019 Published online 28 AUG 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Recent Contributions of Theory to Our Understanding of the Atlantic Meridional Overturning Circulation

Helen L. Johnson<sup>1</sup>, Paola Cessi<sup>2</sup>, David P. Marshall<sup>3</sup>, Fabian Schloesser<sup>4</sup>, and Michael A. Spall<sup>5</sup>

<sup>1</sup>Department of Earth Sciences, University of Oxford, Oxford, UK, <sup>2</sup>Scripps Institution of Oceanography, University of California, San Diego, La Jolla, CA, USA, <sup>3</sup>Department of Physics, University of Oxford, Oxford, UK, <sup>4</sup>International Pacific Research Center, Honolulu, HI, USA, <sup>5</sup>Woods Hole Oceanographic Institution, Woods Hole, MA, USA

**Abstract** Revolutionary observational arrays, together with a new generation of ocean and climate models, have provided new and intriguing insights into the Atlantic Meridional Overturning Circulation (AMOC) over the last two decades. Theoretical models have also changed our view of the AMOC, providing a dynamical framework for understanding the new observations and the results of complex models. In this paper we review recent advances in conceptual understanding of the processes maintaining the AMOC. We discuss recent theoretical models that address issues such as the interplay between surface buoyancy and wind forcing, the extent to which the AMOC is adiabatic, the importance of mesoscale eddies, the interaction between the middepth North Atlantic Deep Water cell and the abyssal Antarctic Bottom Water cell, the role of basin geometry and bathymetry, and the importance of a three-dimensional multiple-basin perspective. We review new paradigms for deep water formation in the high-latitude North Atlantic and the impact of diapycnal mixing on vertical motion in the ocean interior. And we discuss advances in our understanding of the AMOC's stability and its scaling with large-scale meridional density gradients. Along with reviewing theories for the mean AMOC, we consider models of AMOC variability and discuss what we have learned from theory about the detection and meridional propagation of AMOC anomalies. Simple theoretical models remain a vital and powerful tool for articulating our understanding of the AMOC and identifying the processes that are most critical to represent accurately in the next generation of numerical ocean and climate models.

#### 1. Introduction

A net northward flow of warm thermocline and intermediate waters occurs in the upper kilometer throughout the Atlantic basin. This is compensated by a net southward flow of colder North Atlantic Deep Water (NADW) at depths between approximately 1 and 3 km. Water is transformed from the upper to the lower isopycnal layer in the high-latitude North Atlantic where strong cooling results in significant buoyancy loss, locally eroding the stratification maintained by various processes. Transformation from the lower to the upper isopycnal layer occurs via wind-driven and tidally driven diapycnal mixing throughout the global ocean interior as well as in the surface mixed layer of the Southern Ocean after adiabatic wind-driven upwelling. This *Atlantic Meridional Overturning Circulation* (or AMOC) is responsible for transporting significant heat northward in both hemispheres and also plays a key role in the uptake and distribution of other important tracers such as carbon and oxygen. As such, the AMOC is a fundamental component of the climate system.

Figure 1 illustrates the key processes which determine the strength, structure, and variability of the AMOC. Over the last decade or two, groundbreaking basin-wide observational arrays, such as the RAPID-MOCHA array at 26°N (Cunningham et al., 2007; Srokosz & Bryden, 2015), together with other ocean observations (e.g., Cunningham & coauthors, 2019) and a new generation of ocean and climate models, have provided new and intriguing insights into the mean AMOC and its variability. Theoretical models have also contributed to changing our view of the AMOC, providing a dynamical framework for understanding the new observations and some of the complexity described by comprehensive numerical models. In this paper we review recent advances in the theoretical modeling of the AMOC and the conceptual understanding that has resulted from careful, pared-down thinking about the circulation. Readers also interested in a more general review of the AMOC are referred to Buckley and Marshall (2016).



Figure 1. Schematic illustrating the key processes which determine the strength, structure, and variability of the Atlantic Meridional Overturning Circulation. The net northward flow of warm water in the upper kilometer is indicated in red, with the compensating southward flow of colder North Atlantic Deep Water shown by the blue arrow, which deepens along the western boundary. Transformation between these light and dense flows occurs in the subpolar North Atlantic, where buoyancy is lost, and in the Southern Ocean, where buoyancy is gained. Adiabatic flow is possible along isopycnals, illustrated on the vertical section, which outcrop in both regions. In the Southern Ocean, strong westerly winds drive surface Ekman transport and upwelling which steepens isopycnals; the resultant baroclinic instability leads to eddy bolus fluxes which oppose the wind-driven circulation such that the residual overturning circulation, shown in black, is primarily directed along isopycnals in the ocean interior but across isopycnals in the surface mixed layer. A fraction of the upwelled water becomes denser, rather than lighter, in the Southern Ocean, forming Antarctic Bottom Water which subsequently upwells diapycnally to middepths in the Indian and Pacific Oceans. Diapycnal mixing in the Atlantic Ocean interior (dark blue) may also play a role in the water mass transformation. The short extent of the African continent compared to the South American continent promotes import of salty subtropical water from the Indian Ocean as thermocline and intermediate waters return to the Atlantic from the east (red). Mesoscale eddies (white) and wind stress also play a key role in the dynamics and buoyancy budget of the subpolar North Atlantic, with eddies connecting regions of net buoyancy loss to regions of net downwelling. Adjustment of the overturning to changes in wind stress and buoyancy fluxes occurs via boundary and Rossby waves, indicated in pale blue. The locations of the RAPID and Overturning in the Subpolar North Atlantic Programme (OSNAP) observational arrays are marked with black lines.

We consider "recent" advances to include those since the beginning of the AMOC's modern instrumental record, that is, since roughly the start of this century, although we discuss older literature where necessary to demonstrate or motivate a more recent advance. We define "theoretical contributions" broadly, to include analytical theories, conceptual models, and a subset of idealized numerical calculations. We restrict our attention to what is often termed the "upper cell" (more accurately termed the "middepth cell"), that is, the cell whose northward branch transports thermocline and intermediate waters and whose southward branch transports NADW. We discuss the circulation of the "abyssal cell," associated with Antarctic Bottom Water (AABW) formed in the Southern Hemisphere, only where it is important in understanding the circulation of NADW. The AMOC sometimes refers to an overturning in density space via a diapycnal velocity and sometimes it refers to an overturning in depth space, where it represents the Eulerian vertical velocity (Figure 2). We discuss the AMOC in both density and depth space and strive to make connections between the two.

We begin by reviewing the contributions that theoretical models have made to our understanding of the time-averaged AMOC, focusing on the insight the models have provided, individually and collectively, rather than on their mathematical details. Section 2 describes the inclusion of Southern Hemisphere processes, in particular winds, in theoretical models of the AMOC. Section 3 considers the move away from zonally averaged single-basin models to incorporate more complex spatial structure in forcing and circulation. Section 4 revisits the relationship between meridional density gradients and the AMOC, while section 5 reviews recent theoretical progress concerning AMOC stability. Section 6 summarizes current thinking on why we



**Figure 2.** (a) The time-mean residual overturning circulation, zonally integrated over the Atlantic sector, as a function of latitude and  $\sigma_2$ , calculated using the Eulerian plus eddy bolus velocities from the Estimating the Circulation and Climate of the Ocean version 4 state estimate (release 3). South of 33°S, the integral is taken over all longitudes. Notice that at middepths the flow is largely along constant  $\sigma_2$  surfaces. (b) The time-mean meridional overturning circulation, zonally integrated over the Atlantic sector, as a function of latitude and depth, calculated using the Eulerian velocities from the Estimating the Circulation and Climate of the Ocean version 4 state estimate (release 3). South of 33°S, the integral is taken over all longitudes. Notice the very strong clockwise Ekman cell south of 33°S, which is largely compensated by eddy volume fluxes in the residual framework, and the weaker Northern Hemisphere maximum relative to the residual framework in (a).

have an AMOC at all, rather than a Pacific meridional overturning circulation (PMOC), before section 7 reviews theoretical work on NADW formation and the downwelling limb of the AMOC.

We then move on to review models of AMOC variability and consider what we have learned from theory about the detection and meridional propagation of AMOC anomalies. Section 8 focuses on the value that theoretical studies have added to the monitoring and understanding of AMOC variability at 26°N. Section 9 considers recent advances in our understanding of what drives AMOC variability more generally, and section 10 summarizes alternative frameworks for thinking about the AMOC which provide additional insight. The paper closes in section 11 with a summary of the major contributions that theory has made over the last decade and some speculation about what it may contribute in the next decade.

## 2. Southern Hemisphere Processes

A key ingredient in almost all theoretical models of the global-scale AMOC developed over the last two decades is the eastward wind stress over the Southern Ocean. Marshall and Speer (2012) provide a helpful review of the crucial role the Southern Ocean plays in the AMOC. Its unique importance arises from the absence of meridional continental barriers, requiring that the input of momentum by the wind stress at the surface is balanced by the bottom form stress across abyssal topographic obstacles (Munk & Palmén, 1951). The surface meridional Ekman layer transport is compensated by geostrophic return flow at depth, such that the wind-driven Ekman overturning cell extends down to the abyss, upwelling cold, dense water to the south, and downwelling warm, buoyant water to the north, creating strong lateral density gradients over the full fluid column. The resultant baroclinic instability generates an intense baroclinic eddy field and eddy-induced isopycnal slumping which opposes the Eulerian mean isopycnal steepening (Danabasoglu et al., 1994). In equilibrium, the net "residual circulation," including both Eulerian-mean and eddy-induced components (Gent et al., 1995), is primarily directed along isopycnals in the ocean interior but across isopycnals within the surface mixed layer, the latter at a rate set by surface buoyancy fluxes (Marshall, 1997; Speer et al., 2000; Marshall & Radko, 2003). This net residual circulation, together with diapycnal mixing over the global ocean, balances the formation of deep water in the North Atlantic (Figure 1).

The impact of Southern Ocean winds on the AMOC was highlighted by Toggweiler and Samuels (1995) in a series of integrations with an ocean general circulation model (GCM). The first theoretical model to include a pivotal role for Ekman transport in the Southern Ocean around Antarctica in setting the mean strength of the AMOC was Gnanadesikan (1999). His simple and elegant box model explores the relative roles of NADW formation at high latitudes in the North Atlantic, Southern Ocean Ekman transport, Southern Ocean eddies, and diapycnal mixing. Many subsequent models have followed up on these ideas and in particular the extent to which the AMOC is directed along isopycnals, such that no water mass transformation (and hence diapycnal mixing) is required within the ocean interior.

The potential for an adiabatic pole-to-pole overturning circulation (Figure 3a) is most clearly illustrated by Wolfe and Cessi (2011) using a zonally averaged, analytical model, in combination with both coarse resolution and eddy-resolving numerical simulations. They show that, in a single ocean basin connected to a reentrant channel, an adiabatic pole-to-pole overturning circulation is possible provided there is (a) a thermally indirect, wind-driven overturning circulation in the reentrant channel and (b) a set of isopycnals which outcrop in both the reentrant channel and the Northern Hemisphere. In the limit of weak interior diapycnal mixing, the net overturning circulation must be dominated by this adiabatic cell. An important, and as yet poorly understood, component of this pole-to-pole overturning cell is the transformation of potential vorticity that must take place in the northward flowing limb to convert the negative potential vorticity coming from the Southern Hemisphere into the positive potential vorticity found in the Northern Hemisphere (and vice versa for the deep southward flowing limb). In GCMs, this transformation occurs in narrow laminar viscous boundary layers where vorticity can be modified adiabatically (Edwards & Pedlosky, 1998), whereas turbulent processes are likely to play a role in the ocean.

Radko and Kamenkovich (2011) also develop an analytical model that allows for adiabatic solutions. Their model combines classical elements of large-scale circulation theory, including simple models of the thermocline, inertial western boundary current, and eddy-controlled Antarctic Circumpolar Current (ACC) in a 2.5 layer framework to predict the stratification and AMOC transport as a function of surface forcing. Inclusion of possible diabatic effects substantially influences the AMOC transport but not the stratification.

The isopycnal connection between the Southern Ocean and the high-latitude North Atlantic links the surface buoyancy distribution at these end points of the AMOC. This implies that, as well as the Southern Hemisphere westerlies, the Northern Hemisphere winds are also important (see sections 8 and 9 and Cessi, 2018). In addition, in the purely adiabatic case there can be no net source or sink of buoyancy within the volume enclosed by the isopycnals connecting the AMOC end points. Since buoyancy is lost at the isopycnal outcrops in the North Atlantic (corresponding to NADW formation; see section 7), either there is diapycnal mixing along the path of the AMOC or buoyancy must be gained in the upwelling region of the Southern Ocean, or both. The contribution of diapycnal mixing is increased if the path of the lower branch of the overturning is extended, and this can be achieved by connecting the return flow of the AMOC to the abyssal AABW cells in the Indian and Pacific in what has become known as a "figure-of-eight" loop (see Figure 1, section 3, and Talley, 2013). (a)





**Figure 3.** (a) A sketch of the meridional overturning circulation (MOC) from Wolfe and Cessi (2011). A pole-to-pole cell (thick solid line) with sinking in high latitudes in the North Atlantic and upwelling in the Antarctic Circumpolar Current (ACC) region coexists with weaker diffusive cells characterized by high-latitude sinking in each hemisphere and upwelling mostly confined to the same hemisphere (thick dashed lines). The thin solid lines show isopycnals. The isopycnals in the ventilated thermocline region do not outcrop in the ACC region. The isopycnals in the heavily shaded region outcrop in the channel but not in the North Atlantic. The group of three intermediate isopycnals outcrop both in the ACC and the North Atlantic, and it is along these surfaces that an adiabatic pole-to-pole residual overturning circulation can exist, with water mass transformation confined to the mixed layer. (b) Zonally averaged sections of residual circulation (Sv; blue/red) and temperature (°C, black lines) from (left) the general circulation model simulations and (right) theory in Nikurashin and Vallis (2012). The three rows show (top) the control experiment with  $\kappa_{\nu} = 2 \times 10^{-5} \text{ m}^2/\text{s}$  and  $\tau = 0.2 \text{ N/m}^2$ , (middle) an enhanced mixing experiment with  $\kappa_{\nu} = 5 \times 10^{-5} \text{ m}^2/\text{s}$ , and (bottom) a reduced Southern Ocean wind experiment with  $\tau = 0.1 \text{ N/m}^2$ .

Shakespeare and Hogg (2012) develop a three-layer extension of Gnanadesikan (1999) to include the abyssal cell showing that, in the adiabatic limit, the upper cell's overturning circulation scales linearly with both Southern Ocean wind stress and Northern Hemisphere buoyancy loss. They note, however, that buoyancy loss around Antarctica is also important because of the role it plays in setting the stratification; the upper and abyssal cells are coupled due to the requirement for thermodynamic equilibrium.

Although analytically tractable only in the adiabatic limit, the model of Nikurashin and Vallis (2012) is notable as the first theory for the zonally averaged stratification and overturning circulation, as continuous functions of depth and latitude, that includes both adiabatic and diabatic components. The model matches solutions in three regions—the circumpolar channel, northern isopycnal outcrop region, and the basin in between—and agrees well with the results of three-dimensional numerical simulations in a single-basin domain with a reentrant channel to the south (Figure 3b). The upper cell overturning circulation scales linearly with the Southern Ocean westerly winds in the weak diapycnal and eddy mixing limit, while for strong diapycnal mixing (or weak winds) the overturning scales as  $\kappa^{2/3}$ , where  $\kappa$  is the coefficient of diapycnal mixing, in line with many previous studies (Welander, 1971).

Before moving on from zonally averaged models, we note an important caveat in our theoretical understanding of the AMOC, which is the reduced sensitivity of both the slope of the density surfaces and the residual overturning in the Southern Ocean to the surface wind stress in models with explicit, rather than parameterized eddies, known, respectively, as "eddy saturation" and "eddy compensation" (e.g., Tansley & Marshall, 2001; Hallberg & Gnanadesikan, 2001, 2006; Farneti et al., 2010; Viebahn & Eden, 2010; Abernathey et al., 2011; Morrison & Hogg, 2013; Munday et al., 2013). Both eddy saturation and eddy compensation may impact the relationship between the AMOC and Southern Hemisphere winds: the slope of density surfaces across the Southern Ocean is important in determining the equilibrium stratification of the Atlantic (following Gnanadesikan, 1999), while the residual overturning in the Southern Ocean must equal the strength of the AMOC in the limit of an adiabatic pole-to-pole circulation (Wolfe & Cessi, 2011). As a consequence, the linear relationship between the AMOC and the strength of Southern Hemisphere winds exhibited by models such as Shakespeare and Hogg (2012) and Nikurashin and Vallis (2012) may overestimate the sensitivity of the AMOC, especially under strong winds (see, e.g., Radko & Kamenkovich, 2011; McCreary et al., 2016). Preliminary investigations with a new eddy closure, in which the eddy diffusivity varies in proportion to the eddy energy field, suggest that it may be possible to capture the effects of eddy saturation and eddy compensation without explicitly resolving the eddies (Mak et al., 2018), although the robustness of these results and detailed implications for the sensitivity of the AMOC to Southern Hemisphere wind stress requires further investigation.

#### 3. From Two Dimensions to Three Dimensions to Multiple Basins

Samelson (2009) and Bell (2015) take a different approach to including the impact of surface wind stress, allowing the stratification to vary zonally as well as meridionally in response to the wind forcing, which then in turn allows for zonal asymmetry in the effectiveness of surface buoyancy fluxes. Bell (2015) balances the water mass transformation due to surface heat loss in the Northern Hemisphere with that due to the wind-driven shoaling of isopycnals in the south west Atlantic, just north of Drake Passage, where heat is gained by the ocean (Figures 1 and 4). The strength of the overturning circulation in this model depends on the range of wind stresses over the Southern Ocean, and the southernmost latitude at which heat loss occurs in the North Atlantic, consistent with both a linear dependence on Southern Hemisphere winds and with the importance of a shared range of isopycnal outcrops between hemispheres.

Although Bell (2015) and the reduced-gravity model of Samelson (2009) retain a simple layered structure in the vertical and are restricted to a single basin, the move away from a zonally averaged picture permits the wind stress to play a more complex role in the dynamics of the overturning circulation, an emerging theme which is also reflected in some theoretical models of AMOC variability (see sections 8 and 9), and appears to be increasingly borne out by observations and GCM simulations (e.g., Polo et al., 2014; Williams et al., 2014; Pillar et al., 2016). To extend these kinds of theoretical solutions to the full global ocean will require a three-dimensional matching condition between the circulations in the Southern Ocean and in the Atlantic: Marshall et al. (2016) and McCreary et al. (2016) have recently extended ideas of Gill (1968) to explore how a dynamically consistent solution for the Southern Ocean and Atlantic basin affects the stratification, overturning, and ACC.





**Figure 4.** Schematic illustrating the regions in which there is a nonzero air-sea heat flux  $Q_H$  in the two-layer isopycnal model of Bell (2015). The ocean occupies a rectangular basin spanning the equator with a periodic channel at its southern boundary. The lower layer shoals or outcrops in the diagonally shaded regions. The zonal wind stress  $\tau_x(y)$  is shown on the left of the figure. Heat loss in the north of the basin ( $Q_H < 0$ ) is balanced by heat gain in the southwest of the basin ( $Q_H > 0$ ) where wind-driven shoaling of isopycnals brings dense water to the surface.

The move away from zonally averaged models is symptomatic of a trend over the last decade from two-dimensional to three-dimensional AMOC theories, and it is becoming increasingly clear that models need to consider more than one basin. Evidence from hydrographic observations suggests that much of the NADW that upwells in the Southern Ocean, rather than being directly converted to thermocline and intermediate waters by buoyancy input at the surface, is exported to depth as AABW, from where it must subsequently upwell diffusively, mostly in the Indo-Pacific, to rejoin the upper cell (Figure 1; Schmitz, 1995; Talley, 2013).

Thompson et al. (2016) present a multibasin residual-mean model of the global overturning circulation that involves zonal mass transport between basins via the ACC and the Indonesian Throughflow (Figure 5). The model is formulated as a two-basin, four-layer, box model. A portion of the NADW that flows into the Southern Ocean is returned directly upward and northward in the Atlantic sector to close the AMOC cell, while another portion produces a convergence of abyssal waters, which flows into the Indo-Pacific sector of the Southern Ocean. From the abyssal Indo-Pacific this water upwells diapycnally to eventually rejoin the upper cell. This model finds that closure of the overturning cell directly in the Atlantic versus closure through the Indo-Pacific abyss depends on the amplitude of the overturning relative to the deep diffusivity, with the former prevailing for large overturning (or weak diffusivity). The results suggest a minimal role for a closed overturning cell in the present-day Atlantic alone and have implications for the distribution and residence time of tracers, as well as the pathways and sensitivity of the overturning to changed forcings (e.g., see Ferrari et al. (2014) who consider how the circulation may have worked under glacial boundary conditions).





**Figure 5.** Two-dimensional and three-dimensional depictions of the meridional overturning circulation from Thompson et al. (2016). The schematic on the left shows the zonally averaged overturning in depth-latitude space. The green and blue curves are typically viewed as distinct overturning cells associated with North Atlantic Deep Water formation and Antarctic Bottom Water formation, respectively. The idealized three-dimensional schematic on the right illustrates the single figure-of-eight overturning circulation of Talley (2013). Here, the overturning cycles through both the Atlantic and Pacific Basins, either through the Antarctic Circumpolar Current (ACC) or the Indonesian Throughflow, before closing. Rather than two distinct cells, the overturning more closely approximates a single figure-of-eight loop.

Other studies have addressed the same question using a reduced-gravity framework. Building on the work of Johnson and Marshall (2004) and Allison (2009), Jones and Cessi (2016) present a two-basin version of the Gnanadesikan (1999) model which shows that a difference in layer depth between the basins leads to a geostrophically balanced exchange flow from the Pacific-like (nonsinking) basin to the Atlantic in the surface layer, compensated at depth. Ferrari et al. (2017) use a similar two-layer, two-basin model to illustrate that, in the present-day climate, the overturning circulation is best described as the combination of three circulations: an adiabatic overturning circulation in the Atlantic Ocean, associated with transformation of thermocline and intermediate waters to deep waters in the north, a diabatic overturning circulation in the Indo-Pacific Ocean, associated with transformation of abyssal to deep waters by mixing, and an interbasin circulation that exchanges waters geostrophically between the two basins through the Southern Ocean.

These models reflect our increasing appreciation that, even if an adiabatic overturning is possible within the Atlantic basin, water mass transformation in the Indo-Pacific plays an important role in the buoyancy budget of the global overturning circulation (e.g., Newsom & Thompson, 2018) and impacts indirectly on the overall strength of the AMOC.

#### 4. Does the AMOC Scale With Meridional Density Gradients?

It is clear from the papers reviewed in section 2 that the overturning circulation depends strongly on Southern Hemisphere wind stress, as well as diapycnal mixing. However, surface buoyancy loss in the North Atlantic is critical for the formation of NADW, and earlier conceptual models of a thermohaline circulation assumed that the overturning should depend on the large-scale meridional density gradient. Many authors (beginning with Wright & Stocker, 1991; Wright et al., 1995, and more recently Wolfe & Cessi, 2010; Cimatoribus et al., 2013) have sought to reconcile this with the fact that meridional transports must be in thermal wind balance with zonal, rather than meridional, density differences.

In reduced-gravity models, fast boundary waves act to remove pressure gradients along the equator and along the eastern boundary (e.g., Cessi & Louazel, 2001; Johnson & Marshall, 2002a; Marshall & Johnson, 2013, 2017; Bell, 2015;). When buoyancy fluxes force the layer interface to outcrop along the eastern boundary, the resulting pressure gradient drives a convergence of flow toward this boundary which is balanced by deep water formation. Rossby waves propagate the eastern boundary pressure westward across the basin such that the deep water formation region is zonally connected to the western boundary. The strength of the overturning circulation in such models is proportional to the depth-integrated meridional pressure difference, in turn proportional to  $\Delta\rho H^2$  where  $\Delta\rho$  is the difference between the layer densities and *H* the eastern boundary layer thickness. Assuming that upwelling occurs outside the North Atlantic and that the large-scale circula-





**Figure 6.** Temperature (contours) and vertical velocities (shading) as a function of depth and latitude along (a) eastern, and (b) western boundaries in an idealized general circulation model simulation of the circulation in a rectangular, single hemisphere domain driven by a large-scale surface temperature gradient and winds (Schloesser et al., 2014). The magenta curve indicates the eastern boundary mixed layer thickness derived by Sumata and Kubokawa (2001), and cyan curves indicate analytical solutions for the eastern and western boundary layer depths in a two-layer model.

tion is geostrophic, it follows that zonal and meridional pressure gradients are closely related. This scaling also highlights that the vertical stratification is critical in determining how a meridional density difference  $\Delta \rho$  is converted into a pressure gradient.

While the circulation in GCMs forced by large-scale buoyancy fluxes is more complex, it shares many of the key features described above. In particular, numerical model experiments agree with simple layer models and scaling arguments in finding that the AMOC strength is proportional to the depth-integrated meridional pressure difference  $\Delta\rho H^2$  (e.g., Robinson & Stommel, 1959; Bryan, 1987; Marotzke, 1997; Gnanadesikan, 1999; de Boer et al., 2010). The depth scale *H* is generally interpreted as the pycnocline depth, but several recent studies have made progress in developing more useful scalings. For example, de Boer et al. (2010) choose the density difference at the surface (or integrated over 0-1,400m) between the equator and high latitudes and show across a broad suite of GCM experiments that the AMOC scaling is only appropriate if *H* is interpreted as the depth of the maximum overturning streamfunction rather than the depth of the pycnocline.

Schloesser et al. (2012, 2014) and Butler et al. (2016) base their scalings directly on the meridional pressure gradient rather than the density gradient, showing that meridional overturning should therefore scale with the meridional density difference twice integrated in the vertical. Because isopycnal slopes are relatively weak along the equator, taking the density difference along the eastern or western boundary in a closed basin yields similar results. Butler et al. (2016) retain the depth dependence in  $\Delta\rho$  to obtain a scaling for the meridional overturning streamfunction as a function of depth for each basin, which agrees well with both deterministically and stochastically buoyancy forced GCM experiments on multidecadal and longer timescales. Both de Boer et al. (2010) and Butler et al. (2016) use gradients along the western boundary in their scalings.

The importance of the western boundary in meridional density scalings is also emphasized by Sijp et al. (2012) and is consistent with the results of Marshall and Pillar (2011) who diagnose the rotational component of the forces in the meridional momentum equation, that is, the nondivergent part of each force, which projects directly onto the acceleration of the flow. They show that on the western boundary the rotational component of the buoyancy force cannot be compensated by the rotational component of the Coriolis force, which vanishes at the wall, and hence, an overturning cell is accelerated along the western boundary.

In response to a large-scale meridional density gradient, water sinks and the thermocline deepens along the eastern boundary (see Sumata & Kubokawa, 2001, and Figure 6). The unstable eastern boundary isopycnal structure is quickly homogenized by mixing processes as it is propagated westward by Rossby waves (Schloesser et al., 2012), and the resulting flow convergence and water mass transformation are similar to that in simple isopycnal models which do not include these eastern boundary processes. By documenting how instability processes erode the eastern boundary density structure and lead to northward convergence in an eastern boundary current in an eddy-resolving GCM, Cessi and Wolfe (2013) reconcile the opposing requirements of a meridional gradient in buoyancy at the surface and no flow normal to the boundary, and highlight the importance of ageostrophic eastern boundary processes (Figure 1).

As a consequence of these boundary processes, different circulations develop in response to surface density gradients along lateral boundaries compared with in the ocean interior (see also section 7). Pedlosky and Spall (2005) explore how buoyancy-forced dissipation of Rossby waves as they travel westward across the basin creates convergence of poleward surface flow in the North Atlantic, away from the lateral boundaries. Although overturning circulation driven by density gradients in the interior of an ocean basin also scales with the meridional pressure gradient, it is substantially weaker than that associated with density gradients along lateral boundaries (e.g., Schloesser, 2015). Hence, it is the density gradient along the boundaries that dominates in setting the strength of the large-scale AMOC (Spall & Pickart, 2001).

Exploiting the relative flatness of the isopycnals along the eastern boundary, equatorward of regions of deep water formation, Marshall and Johnson (2017) derive a simple relationship based on thermal wind balance for the relative strengths of the AMOC and the ACC (excluding contributions to the latter from bottom flow). This relationship involves the ratios of three depth scales—the depth of the maximum AMOC streamfunction, the depth of the ACC, and the *e*-folding depth of the stratification in the Atlantic—and explains the  $8 \pm 2$  ratio between the mean ACC and AMOC volume transports.

While surface buoyancy fluxes alone are sufficient for generating a meridional overturning circulation in idealized model experiments (e.g., Gjermundsen & Lacasce, 2017), large-scale wind forcing allows for the lateral separation of the northward and southward flows. For example, in the subpolar region of the Northern Hemisphere, the northward, surface branch of the AMOC occurs in the interior of the subpolar gyre, while the southward return flow occurs along the western boundary. Because water is cooled as it flows northward in the gyre, wind forcing provides an additional mechanism for convergence of the surface meridional flow in the North Atlantic (Schloesser et al., 2014). This, together with the importance of meridional Ekman transport for the AMOC (see section 8), impacts the applicability of scalings based on the meridional density gradient in realistic numerical simulations and the ocean.

#### 5. AMOC Stability

The debate about whether meridional density gradients set, or merely reflect, the strength of the AMOC has important implications for our understanding of its stability, which depends on thermohaline feedbacks. Northward advection of salt by the overturning circulation results in the amplification of overturning anomalies and the possibility of abrupt change between bistable regimes (the salt advection feedback identified by Stommel (1961)). Many studies over the past decade have attempted to reconcile the existence of multiple equilibria with a mechanically rather than buoyancy forced AMOC. Several authors have tackled this problem using box models, beginning with Johnson et al. (2007) who incorporated salt advection into the Gnanadesikan (1999) model using a Stommel (1961) formulation. Fürst and Levermann (2012), Cimatoribus et al. (2014), and others have followed up, and it is clear that multiple equilibria and abrupt transitions between them are possible, at least in a box model framework, in both wind- and mixing-driven regimes.

More recently, Wolfe and Cessi (2014, 2015) have explored the role of salt advection in setting the range of isopycnal outcrops that are common to both hemispheres, showing that even in an adiabatic pole-to-pole overturning circulation, salt advection can lead to hysteresis and oscillatory behavior in an idealized numerical simulation of a single basin with a reentrant channel.

An interesting but controversial suggestion has been that the northward freshwater transport across  $30^{\circ}$ S by the meridional overturning circulation, commonly referred to as  $M_{ov}$ , acts as an indicator of whether the AMOC is in a bistable regime, that is, a regime in which both a strong, thermally driven overturning circulation and a weakened or reversed circulation are stable. Based on a simple model, de Vries and Weber (2005) argue that multiple equilibria can only exist when the overturning circulation exports freshwater

from the Atlantic basin, that is, when  $M_{ov} < 0$  and the upper, incoming branch of the overturning is saltier than the NADW that leaves the Atlantic. If  $M_{ov} > 0$ , the feedback is damping and the AMOC is monostable.

This idea has been proposed, fleshed out, explored, and tested in a range of theoretical, idealized, and more realistic models (see discussion in Drijfhout et al. (2013)). Refer to Dijkstra (2007), Huisman et al. (2010), and Cimatoribus et al. (2012, 2014) for evidence of the role of  $M_{ov}$  in stability; Weber et al. (2007), Drijfhout et al. (2011), Hawkins et al. (2011), Weaver et al. (2012), Mecking et al. (2016), and Gent (2018) for  $M_{ov}$  diagnostics in climate models; and Liu and Liu (2013), Liu et al. (2013), and Sijp (2012) for other, related, stability indicators. Rather worryingly,  $M_{ov}$  is positive in many climate models, yet negative when diagnosed from observations (Liu & Liu, 2014; Mecking et al., 2017), suggesting that many climate models have a fresh South Atlantic thermocline bias and may not correctly represent the stability of the AMOC. The horizontal gyre component of freshwater transport into the Atlantic is also important: some models suggest that it is of the opposite sign to, and larger in magnitude than, the overturning component (Gent, 2018). Evidence in support of the mechanism in GCMs is inconclusive, and the importance of  $M_{ov}$  remains under debate.

The stability of the AMOC is the subject of another review paper in this special collection, so we refrain from further discussion here and refer the reader to Weijer et al. (2019).

#### 6. Why Does Deep Water Form in the North Atlantic But Not the North Pacific?

Related to the ongoing debate about the stability of the AMOC is the question of why we have an AMOC, rather than a PMOC, in the first place. It is well established that the North Pacific is about 2 psu fresher than the North Atlantic, and this small difference prevents deep water formation in the Pacific (Warren, 1983). Therefore, to understand why we have an AMOC instead of a PMOC, the higher salinity of the Atlantic over the Pacific must be explained.

A recent review summarizes the state of knowledge on the Atlantic-Pacific salinity difference (Ferreira et al., 2018), and its conclusions are summarized here. Atmospheric and oceanic processes both contribute to the asymmetry in the salinity distribution, in approximately equal proportions. The main reasons that atmospheric processes favor more precipitation in the Pacific than in the Atlantic sector are that (i) evaporation occurs mainly in coastal regions which represent a larger fraction of the area in a narrow basin (Schmitt et al., 1989), (ii) there is a fetch of about 3,000 km before coastal evaporation precipitates, penalizing precipitation in the narrow Atlantic sector in favor of the wide Pacific Sector (Ferreira et al., 2010), and (iii) the monsoonal circulation localizes moisture convergence in the Pacific (Emile-Geay et al., 2003).

The main oceanic processes favoring Atlantic salinification are (i) the advection of salt-rich subtropical waters northward by the AMOC itself, which maintains the preference for Atlantic sinking—this is the salt advection feedback (Stommel, 1961), and (ii) the relatively short meridional extent, and hence subtropical termination, of the African continent east of the Atlantic, which promotes import of salty subtropical water from the Indian Ocean (Reid, 1961; Gordon, 1986; Nilsson et al., 2013; Cessi & Jones, 2017). Because of the massive wind-driven subtropical gyre of the Southern Hemisphere (the "supergyre" of Ridgway & Dunn, 2003; Speich et al., 2007), thermocline and intermediate water flows westward around the tip of South Africa, that is, from the Indian to the Atlantic, rather than returning to the Atlantic eastward from the Pacific around South America. This path around the supergyre results in an interbasin transport of salt in the Southern Ocean, from the Indo-Pacific toward the Atlantic (see Figures 7 and 1 and Cessi & Jones, 2017; Nilsson et al., 2013). The dense outflow from the highly evaporative Mediterranean Sea has previously been considered important for the preference of Atlantic over Pacific sinking (Reid, 1979). However, recent studies (Blanke et al., 2006; Jia et al., 2007) do not conclusively support the notion that the outflow from this semienclosed area contributes to the salt budget of the upper limb of the AMOC beyond the salinification that would be effected by a freshwater loss of about 0.05 Sv over a comparable area of the Atlantic.

It remains unclear which, if any, of the three atmospheric and two oceanic mechanisms listed above is most important and whether any are necessary rather than simply contributing. The multiplicity of causes may reflect an intrinsic complexity of the system or simply our lack of understanding to date.

In addition to having higher salinity, the North Atlantic is more connected to the Arctic than the North Pacific is, and this facilitates the formation of deep water as discussed in the following section.





**Figure 7.** The time-mean surface salinity in a numerical model of the ocean circulation in two idealized basins separated by narrow continents (in gray) and connected by a circumpolar channel to the south. The model is forced at the surface by prescribed zonally uniform wind stress, freshwater flux, and temperature relaxation (Cessi & Jones, 2017). The dynamics spontaneously generate a zonally asymmetric circulation, with a meridional interhemispheric overturning in the narrow basin and no overturning in the wide basin. Despite zonally uniform surface freshwater flux, the surface salinity near 70°N is almost 2 psu larger in the narrow basin, where sinking takes place, than in the wide basin where there is no sinking. The subtropical interbasin connection at 35°S is crucial for the salinification of the narrow basin and the localization of the overturning, because it allows transport of salty subtropical waters from the wide to the narrow basin.

## 7. Deep Water Formation and the Downwelling Limb of the AMOC

Buoyancy loss at high latitudes in the North Atlantic and Nordic Seas results in both a densification of near-surface waters and a net downwelling from the upper ocean to middepths. This transformation connects the northward flowing upper limb of the AMOC to the southward flowing lower limb of the AMOC. While the buoyancy loss takes place over large areas of the ocean, the dominant water masses are formed in relatively isolated regions of deep convection found in the Labrador Sea, Mediterranean Sea, Greenland Sea (Marshall & Schott, 1999), and more intermittently, the Irminger Sea (Pickart et al., 2003; de Jong & de Steur, 2016). The properties of the waters formed in these regions determine the isopycnals that connect the downwelling limb of the AMOC to the outcropping regions in the Southern Ocean.

It is important to distinguish between water mass transformation and Eulerian downwelling. Both are important, and although connected, they are distinct in magnitude, location, and physics. Send and Marshall (1995) found that the net downwelling in a nonhydrostatic model of buoyancy-forced convection was nearly zero, suggesting that the dominant regions of water mass transformation are not regions of net sinking. This was also inferred from the low-resolution basin-scale models of the AMOC by Marotzke and Scott (1999) and Spall and Pickart (2001), where it was shown that regions of deep convection are found in the basin interior while regions of net sinking are found near the boundaries. The localization of downwelling near side walls was anticipated by the linearized theories of Barcilon and Pedlosky (1967), Spall (2003), and Pedlosky (2003) and is consistent with the theoretical work by MacCready (1994) on the slow decay and descent of deep boundary currents as they travel around a basin. For stratified flows, the downwelling width scales with the baroclinic deformation radius times the square root of the horizontal Prandtl number. The buoyancy loss, due to direct cooling by the atmosphere or lateral eddy fluxes into the interior, is balanced by downward advection of the mean stratification (the downwelling limb of the AMOC) and alongstream advection in narrow boundary currents. The downwelling produces stretching of planetary vorticity, which is balanced by lateral diffusion of vorticity into the boundary. The concentration of downwelling near the boundaries is supported by observations (Pickart & Spall, 2007), in laboratory experiments (Cenedese, 2012), and in eddy-permitting GCMs (Waldman et al., 2018).

Since the large-scale upper ocean flows are in geostrophic balance to leading order, the net inflow to/outflow from a marginal sea at any given depth, which is an indication of up/downwelling within the marginal sea, is proportional to the pressure change across the opening. The pressure field is, to a good approximation, hydrostatic, and therefore, the net downwelling is indirectly determined by the density field. To understand





**Figure 8.** Schematic illustrating the key circulation features operating in North Atlantic marginal seas where dense water formation occurs: (a) warm inflowing boundary current, (b) buoyancy loss to the atmosphere, (c) convective plumes, (d) buoyant eddies shed from the boundary current, (e) cooler outflowing boundary current, and (f) downwelling along the topography.

what determines the net downwelling, one must first understand the buoyancy budget of the marginal sea. Regions of deep convection in the North Atlantic share some common characteristics: They have weak mean flows in the interior, doming isopycnals and are surrounded by a cyclonic boundary current (Marshall & Schott, 1999). Heat loss in the basin interior is balanced by lateral eddy fluxes from the boundary current. This results in an increase in density along the boundary and a barotropization of the boundary current as it flows cyclonically around the basin. In order to remain in geostrophic balance with this reduced vertical shear, downwelling is required near the boundary, from the upper layers to the depths of deep convection. This is the downwelling limb of the AMOC that is forced by surface buoyancy fluxes. The key processes are illustrated in Figure 8. Buoyancy loss by deep convection in the interior (diapycnal AMOC) and vertical transport near the boundary (Eulerian AMOC) are connected through lateral eddy fluxes. The temperature and salinity of the convective water mass and the waters that are exported from the basin depend fundamentally on the balance between mean advection, eddy fluxes, and air-sea exchange (Spall, 2004, 2012; Straneo, 2006). Mesoscale eddies play two important roles in the downwelling limb of the AMOC. First, waters are able to downwell, while isopycnals are rising (following the mean flow cyclonically around the basin), even in the limit of weak diapycnal mixing, through the action of mesoscale eddies, as revealed by a Transformed Eulerian Mean formulation (Cessi et al., 2010; Spall, 2010). Eddies play a similarly fundamental role in balancing the upwelling of the lower branch of the AMOC in the Southern Ocean in the presence of downward Ekman pumping (Danabasoglu et al., 1994; Marshall, 1997; Marshall & Radko, 2003). Second, eddies provide the heat and salt to balance air-sea exchange in the basin interior. In regions where the boundary current is very stable, eddies are not able to flux much heat into the interior and the atmosphere strongly cools the basin. This decreases the heat flux into the atmosphere because the mean air-sea temperature difference is reduced. On the other hand, if the boundary current is able to shed warm, salty eddies into the interior, the regions of deep convection are warmer and the resulting air-sea heat flux is increased. So eddies are essential for proper representation of both the Eulerian and the diapycnal components of the AMOC, and air-sea exchange, in regions of deep convection.

Finally, we note that an exception to the gradual descent of dense water within deep boundary currents occurs when dense water exits a marginal sea over a sill and encounters the lighter ambient waters of the Atlantic, resulting in a dense overflow and significant entrainment. The dynamics were first explored in a



numerical streamtube (one-dimensional) model by Price and Baringer (1994). While there has been significant recent progress on improving the representation of overflows and entrainment in ocean and climate models (e.g., Legg et al., 2009) and their impact on the AMOC and its constituent water masses may be significant (e.g., Danabasoglu et al., 2010), this improved understanding has yet to be incorporated into basin-scale theoretical models of the AMOC.

### 8. Monitoring and Understanding Observed AMOC Variability at 26°N

Since 2004 the RAPID-MOCHA array has been measuring the AMOC at 26°N. Profiles of density on eastern and western boundaries of the Atlantic (as well as on either side of the mid-Atlantic ridge) are used to determine the geostrophic ocean interior flow via thermal wind balance. When combined with cable measurements of the flow through Florida Straits, direct current meter measurements on the western boundary, an estimate of the surface Ekman layer transport based on atmospheric reanalysis winds, and a mass conservation constraint, the array provides daily estimates of the strength and vertical structure of the meridional overturning circulation (Cunningham et al., 2007; McCarthy et al., 2015). This unprecedented decade-long record of the AMOC in the subtropical North Atlantic has revealed large amplitude variability on all timescales accessible to date, as well as what appears to be a longer-term trend (Smeed et al., 2014; Srokosz & Bryden, 2015).

Theoretical studies have played a key role in both challenging and defending the ability of an array such as RAPID-MOCHA to capture long-term trends. One important example concerns the potential role of eddies in low-frequency AMOC variability at 26°N. Wunsch (2008) projected sea surface height variability, which reaches a root-mean-square amplitude of 16 cm near the western side of the subtropical Atlantic, onto vertical modes of horizontal velocity to show that eddies alone could result in a 16 Sv root-mean-square transport variability measured at the array. Wunsch (2008) therefore predicted that any signal of longer-term change in the AMOC at the RAPID-MOCHA array should be swamped by eddies. However, Kanzow et al. (2009) used linear wave theory, together with simple reduced-gravity simulations, mooring, and altimetry data to show that, in fact, the eddy energy drops off sharply right at the western boundary. Matching incoming long Rossby waves with reflected short Rossby waves at the western boundary ensures that density anomalies on the boundary are significantly smaller than those associated with the incoming Rossby wave. Kanzow et al. (2009) and Zhai et al. (2010) show that the same is true for nonlinear eddies in reduced-gravity models. Therefore, provided thermal wind balance for the meridional flow is applied all the way across the basin from boundary to boundary, eddies do not dominate the AMOC variability signal. (Clément et al. (2014) use mooring data to show that the variability of the transbasin geostrophic transport attributed to eddies and Rossby waves is estimated to be 2.6 Sv). The role of eddies in generating low-frequency AMOC variability is still under debate, with some eddy-resolving models (e.g., Hirschi et al., 2013; Thomas & Zhai, 2013) suggesting that the eddy contribution to AMOC variability is significant compared to the seasonal cycle at some latitudes.

Theory continues to play a role in refining and improving our strategies for observing the AMOC. For example, Hughes et al. (2013) demonstrate how the geostrophic portion of the meridional overturning circulation can be monitored using only boundary measurements, while others (e.g., Duchez et al., 2014; Frajka-Williams, 2015) lean on established theory such as Sverdrup balance to develop metrics for the overturning circulation that may allow us to reconstruct its variability over a longer time period. Hughes et al. (2018) show that mesoscale energy is suppressed in bottom pressure on the continental slope, such that variations in pressure along boundaries are coherent over large distances, with likely implications for the coherence of the AMOC.

Several theoretical and simple idealized process models have helped interpret the variability in the observed AMOC time series at the RAPID-MOCHA array. Anything which alters the density or pressure on the eastern or western boundary of the North Atlantic, by definition, influences the AMOC, and there has been a growing realization over the last decade that winds may play a large role beyond that due to the local Ekman transport. The significant seasonal cycle in AMOC transport at 26°N appears to be governed by eastern boundary pressure anomalies. Zhao and Johns (2014a) force a simple linear Rossby wave model with the climatological seasonal cycle in wind stress to show that variations in the thermocline depth forced by wind stress curl anomalies on the eastern side of the RAPID-MOCHA array have a dominant effect on the sea-



**Figure 9.** Low-pass filtered time series of variability in the Atlantic Meridional Overturning Circulation (red), Florida Current transport (blue), Ekman transport (black), and upper mid-ocean geostrophic transport (magenta) at 26°N, from Zhao and Johns (2014b). The dashed lines show in situ observations made by the RAPID-MOCHA array between April 2004 and April 2011. The solid lines show the results of a linear two-layer model forced only by observed wind stress. For better visualization, the mean level of the Florida Current, Atlantic Meridional Overturning Circulation, and upper mid-ocean time series has been shifted by 4 Sv each.

sonal cycle of the interior geostrophic transport, consistent with the observations reported in Chidichimo et al. (2010).

However, Yang (2015) insert walls in a two-layer Atlantic model to show that while eastern boundary pressure anomalies are a key contributor to the seasonal cycle in meridional geostrophic transport, which in turn is the largest contributor to the AMOC seasonal cycle at 26°N, the pressure at the eastern boundary is *not* forced mainly by the local wind stress curl at the African coast. Rather, it is the result of basin-wide adjustments to local and remote wind forcing. In their model, Yang (2015) see a meridional redistribution of the water in each layer between the subtropical and subpolar gyres in response to seasonal variability in Ekman pumping, and it is this which dominates the AMOC seasonal cycle.

It is becoming clear that changes in the geostrophic contribution to the AMOC often involve a complex oceanic adjustment to the wind forcing. Zhao and Johns (2014b) show that most of the observed interannual variability in overturning and its components at 26°N can be reproduced in a linear, two-layer model forced only by winds (Figure 9).

#### 9. Generation and Propagation of Wider AMOC Variability

Progress has also been made in the last decade on understanding the role of surface buoyancy fluxes on longer timescales. For example, Zanna et al. (2011) solve a generalized eigenvalue problem in an idealized single-basin ocean GCM to identify the optimal three-dimensional structure of temperature and salinity perturbations that amplify AMOC anomalies. Their study makes it clear that linearized ocean dynamics can give rise to enhanced AMOC variability if overflows, eddies, or deep convection can excite *deep* high-latitude density anomalies in the northern part of the basin, and these impact the AMOC 7.5 years later via thermal Rossby modes (te Raa & Dijkstra, 2002) and boundary waves (Marshall & Johnson, 2013).

Reduced-gravity, 1.5 layer models that allow for zonal, but not vertical, structure in the overturning circulation and stratification are also highlighting the roles of boundary propagation and interior Rossby waves in the adjustment of the AMOC to a change in forcing. Samelson (2011) derives time-dependent analytical solutions for the warm branch of the AMOC in a single-basin plus reentrant channel which includes Southern Ocean Ekman and eddy volume fluxes, as well as diabatic processes to the north. The adjustment timescale is multidecadal to centennial and proceeds via the boundary and interior adjustment mechanisms identified in Johnson and Marshall (2002a, 2002b), with the eastern boundary upper layer thickness proving to be the key variable. Nieves and Spall (2018) present similar reduced-gravity theory for variability in the deep branch of the overturning circulation in a single basin, explicitly assuming that the eastern boundary layer thickness is equivalent to the domain-averaged layer thickness (valid in the low frequency limit), consistent with flat isopycnals on the eastern boundary and Rossby wave propagation into the interior.

Reduced-gravity models such as these provide a simple analytical framework for exploring the different characteristics of AMOC variability generated by wind forcing and surface buoyancy fluxes, and the combined influence of the two. Zhai et al. (2014) build on Johnson and Marshall (2002a) to propose an analytical theory for the interaction of wind stress and buoyancy fluxes associated with the North Atlantic Oscillation in generating AMOC variability. They construct a volume budget for the upper branch of the AMOC assuming fast boundary wave propagation and westward Rossby wave propagation in the ocean interior (these adjustment mechanisms are illustrated in Figure 1). The effect of stochastic wind forcing over the subpolar basin is integrated along Rossby wave characteristics and consequently results in low-frequency AMOC variability in the rest of the basin. In contrast, North Atlantic Oscillation-related surface buoyancy fluxes excite stochastic variability directly on the western boundary, and this is subsequently communicated into the ocean interior.

Surface buoyancy fluxes and wind stress are intrinsically coupled, and as Williams et al. (2014) point out, variations in the wind over the North Atlantic naturally lead to opposing signs of thermal anomalies, Ekman heat convergence, and (via thermal wind) MOC-minus-Ekman volume transport in the subtropical and subpolar parts of the basin.

Boundary waves are clearly important in determining AMOC variability (also see the adjoint sensitivity studies in Czeschel et al. (2010) and Pillar et al. (2016)). However, the meridional propagation of density and AMOC anomalies in climate models occurs on a wide range of timescales. Marshall and Johnson (2013) provide a possible explanation for some of the differences, solving analytically for the propagation of waves along eastern and western boundaries in a reduced-gravity model. They show that, even in this very simple model, on timescales longer than 1–2 months, boundary waves are not Kelvin waves but rather short and long Rossby waves that satisfy a lateral boundary condition. These Rossby waves propagate cyclonically along coastlines with an along-boundary propagation speed of  $cL_d/\delta$ , where *c* is the gravity wave speed,  $L_d$  is the Rossby deformation radius, and  $\delta$  is the frictional boundary current width. Although the waves are likely more complicated with sloping bottom topography rather than vertical sidewalls (Wise et al., 2018; Hughes et al., 2019), the results of Marshall and Johnson (2013) may explain some of the spread in the meridional propagation speed of AMOC anomalies between different models.

#### **10. Alternative AMOC Frameworks**

The AMOC transport can be diagnosed using depth and latitude (at fixed depth) as coordinates, or density and latitude (at fixed neutral or potential density). The mean overturning streamfunction calculated from the Estimating the Circulation and Climate of the Ocean (ECCO) state estimate is shown in Figure 2. In density coordinates, the time average and zonal integral characterizing the climatological AMOC must be performed at constant density, requiring synoptic velocity and density measurements. The density coordinate framework highlights the transformation between waters of different density classes and is thus more representative of the transport of tracers (Andrews & McIntyre, 1978). The difference between the two frameworks is especially important in high latitudes where the isopycnals slope most dramatically. In the Antarctic circumpolar region the main difference between the transport in depth versus density coordinates is due to the large contribution by mesoscale eddies and standing waves to the transport of density (e.g., Doos & Webb, 1994). In the high latitudes of the North Atlantic the difference arises from the (horizontal) transport effected by the subpolar gyre. The first continuous observations of the overturning in density space at high northern latitudes, made over 2014–2016 by the Overturning in the Subpolar North Atlantic Programme (Lozier et al., 2017), suggest that most of the water mass transformation and its variability occur to the east of Greenland (Lozier et al., 2019).

The water mass transformation framework (Walin, 1982) formulates mass budgets for ocean volumes bounded by scalar surfaces (e.g., isopycnals) and relates ocean circulation to processes which result in a change in the position of these surfaces (see Groeskamp et al., 2019, for a comprehensive review). In an AMOC context, it has been applied, for example, to understand the link between surface forcing, diapycnal mixing, and advection of water masses in the North Atlantic (e.g., Speer & Tziperman, 1992), interannual variations of the AMOC (Marsh, 2000), and the formation of dense Nordic Seas waters (Isachsen et al., 2007),



**Figure 10.** The thermohaline streamfunction from Doos et al. (2012). The deep overturning circulation, indicated in red in Figure 2, is shown here by the blue clockwise circulation, which corresponds to water mass conversion from warm water to more saline, then to cold, then to less saline, and back to warm water. The tropical and Antarctic Bottom Water (AABW) cells flow in the opposite direction. Units are in Sverdrups (Sv) with a volume transport of 4 Sv between contours.

as well as to characterize the AMOC in models (e.g., Marsh et al., 2005). Building on Speer (1993), who decomposed the circulation in buoyancy coordinates into its separate contributions from salinity and potential temperature, Doos et al. (2012), Zika et al. (2012), and Groeskamp et al. (2014) define a streamfunction in thermohaline coordinates ( $\theta$  and *S*; see Figure 10) which provides insight into the AMOC's role in global heat and freshwater transports and allows for the study of processes directly affecting the thermohaline circulation (e.g., Hieronymus et al., 2014). Considering the water mass transformation only in  $\theta$  coordinates allowed Evans et al. (2017) to reconcile observed wind-driven changes in the AMOC at 26°N with observed changes in water masses. Recent efforts to combine the overturning circulation in thermohaline coordinates with a box-inverse method (Mackay et al., 2018) are adding value to current observational campaigns.

#### **11. Concluding Discussion**

Several key themes and questions emerge that have influenced those constructing theoretical models of the AMOC over the last decade. Many of these are illustrated in Figure 1.

- It is now clear that the wind, in both hemispheres, plays a prominent role in setting the mean AMOC strength and determining its variability. This includes the interaction between the wind stress and the surface buoyancy distribution, because wind-driven upwelling, and the vertical flux of buoyancy associated with eddies and gyres, brings water to the surface where its density can be transformed by buoyancy fluxes.
- Simplified models are moving from two-dimensional zonally averaged representations to geometries that capture the fundamentally three-dimensional aspects of the circulation. This includes a distinction between the western boundary and the basin interior and a focus on the circulation between multiple basins.
- The degree to which water mass transformation in the ocean interior is important, or whether the circulation in the Atlantic sector is essentially adiabatic, remains an open question.
- Multiple lines of evidence suggest that eddies at high latitudes in both hemispheres are essential to the dynamics of the AMOC.

Some theoretical models have moved beyond a treatment of ocean dynamics to explore the AMOC's role in the coupled climate system. For example, Marshall and Zanna (2014) explore the processes, including the AMOC, that determine ocean heat uptake in a multilayer version of the Gnanadesikan (1999) model, using it to interpret the results of a climate change experiment in a coupled climate model. Zhai et al. (2011) explore the links between AMOC variability and ocean heat content in the thermocline, showing that low frequency variability in the AMOC is required to impact the ocean's heat content at middle and high latitudes. Meanwhile, other authors are beginning to develop theoretical models that include a representation of the biogeochemical processes that interact with the ocean's overturning circulation to determine the impact of AMOC variability on climate (e.g., Goodwin, 2012).

A major recent theoretical advance, whose implications for our understanding of the AMOC are not yet clear, is our changing paradigm concerning the link between ocean turbulent mixing and water mass transformation. It is classically assumed (Munk, 1966; Munk & Wunsch, 1998; Wunsch & Ferrari, 2004) that a vertical advective-diffusive balance is appropriate in the interior, with dense waters transformed into lighter ones as a result of diffusive fluxes. However, turbulent mixing due to breaking internal waves is stronger over rough bottom topography and decays away from the bottom, which means that a parcel of water in the stratified interior mixes more with the water below it than that above, such that mixing in fact converts light waters into denser ones (de Lavergne et al., 2016; Ferrari et al., 2016). Abyssal waters can therefore only rise toward the surface in narrow turbulent boundary layers that develop along continental margins and abyssal ridges. The abyssal meridional overturning circulation is the small residual of diapycnal sinking in the stratified interior and diapycnal upwelling in boundary layers.

The increase in mixing with depth in the stratified interior is at least partially offset by the decrease in mixing area with depth (de Lavergne et al., 2017). In any case, this new conceptual picture is most likely to be relevant below 2,000 m where small-scale mixing driven by internal wave breaking provides the dominant source of energy to lift water masses, and it is as yet unclear how important it is going to be for the AMOC. There are other important implications, however, such as for the horizontal pathways of abyssal ocean flow, the residence time and distribution of tracers, and the relevance of idealized numerical simulations with vertical sidewalls, since it is now clear that sloping boundaries are crucial in allowing ocean basins to remain stratified.

Theoretical and conceptual models continue to play a vital and revealing role in articulating our current understanding of the overturning circulation and helping us to frame first-order questions that remain open about AMOC dynamics. We close by outlining some of the remaining challenges which we anticipate theory will make a significant contribution to solving over the next decade.

- A quantitative understanding of the processes which determine the mean strength of the residual overturning circulation is still lacking. In the Southern Ocean the Eulerian mean and eddy-driven overturning cells appear to be largely powered by local winds, but the interplay between these and the surface buoyancy fluxes and wind stress in the North Atlantic is still being explored, as is the diapycnal mixing along the whole path of the overturning.
- We know from a host of model studies that AMOC variability depends on the background state. Theoretical models have the potential to help us understand this relationship and make sense of the bewildering variety of AMOC variability that is exhibited by state-of-the-art ocean and climate models.
- The continued development of simple models of the coupled ocean-atmosphere system, building on studies such as Lucarini and Stone (2005), will better describe the thermal air-sea interaction, with important repercussions for the representation of AMOC stability.
- Insight from theoretical studies will help us to prioritize the most critical, often subgrid scale, processes requiring improved representation in numerical ocean circulation models and to develop the appropriate physical parameterizations. For example, it is now clear that submesoscale processes such as mixed layer instabilities and eddies can play an important role in restratifying the water column and balancing surface heat fluxes, with dramatic impacts on mixed layer depths (e.g., Boccaletti et al., 2007). The development of a mixed layer eddy parameterization therefore has the potential to significantly improve the representation of the AMOC in models.

By their very nature, theoretical advances are difficult to foresee! However, several of the developments over the last decade have been catalyzed by recent observations such as those made by the RAPID-MOCHA array at 26°N (e.g., our understanding of the mechanisms via which wind forcing dominates the AMOC seasonal



cycle, and of the role of eddies and boundary propagation). The unprecedented amount of data currently being collected by existing and new arrays throughout the Atlantic (e.g., RAPID-MOCHA (McCarthy et al., 2015), Overturning in the Subpolar North Atlantic Programme (OSNAP, Lozier et al., 2017), SAMOC at 34°S (Meinen et al., 2018), MOVE at 16°N (Send et al., 2011), etc. See Cunningham & coauthors, 2019, for a summary) together with long-term global ocean observing systems such as Argo (http://www.argo.ucsd. edu) and satellite altimetry (https://www.aviso.altimetry.fr/en/) mean that there has never been such an exciting time to be thinking about the Atlantic Ocean. This, along with the substantial contributions made by theoretical models to date, provides powerful motivation for equipping the next generation of physical oceanographers with the skills required to use them as part of their toolbox. We can expect that such models will play a major role in understanding the observations collected over the coming decade, as well as in formulating hypotheses for the community to test using these observations alongside numerical ocean and climate models.

#### References

Abernathey, R., Marshall, J., & Ferreira, D. (2011). The dependence of Southern Ocean meridional overturning on wind stress. *Journal of Physical Oceanography*, 41, 2261–2278. https://doi.org/10.1175/JPO-D-11-023.1

- Allison, L. C. (2009). Spin-up and adjustment of the Antarctic Circumpolar Current and the global pycnocline (Ph.D. thesis), University of Reading.
- Andrews, D. G., & McIntyre, M. E. (1978). An exact theory of nonlinear waves on a Lagrangian-mean flow. Journal of Fluid Mechanics, 89(4), 609–646.
- Barcilon, V., & Pedlosky, J. (1967). A unified linear theory of homogeneous and stratified rotating fluids. *Journal of Fluid Mechanics*, 29, 609–621.

Bell, M. J. (2015). Meridional overturning circulations driven by surface wind and buoyancy forcing. *Journal of Physical Oceanography*, 45(11), 2701–2714. https://doi.org/10.1175/JPO-D-14-0255.1

Blanke, B., Arhan, M., & Speich, S. (2006). Salinity changes along the upper limb of the Atlantic thermohaline circulation. *Geophysical Research Letters*, 33, L06609. https://doi.org/10.1029/2005GL024938

Boccaletti, G., Ferrari, R., & Fox-Kemper, B. (2007). Mixed layer instabilities and restratification. *Journal of Physical Oceanography*, 37(9), 2228–2250. https://doi.org/10.1175/JPO3101.1

Bryan, F. (1987). Parameter sensitivity of primitive equation ocean general circulation models. Journal of Physical Oceanography, 17(7), 970–985.

Buckley, M. W., & Marshall, J. (2016). Observations, inferences, and mechanisms of Atlantic Meridional Overturning Circulation variability: A review. *Reviews of Geophysics*, 54, 5–63. https://doi.org/10.1002/2015RG000493

Butler, E. D., Oliver, K. I. C., Hirschi, J. J.-M., & Mecking, J. V. (2016). Reconstructing global overturning from meridional density gradients. *Climate Dynamics*, 46(7), 2593–2610. https://doi.org/10.1007/s00382-015-2719-6

Cenedese, C. (2012). Downwelling in basins subject to buoyancy loss. *Journal of Physical Oceanography*, 42, 1817–1833. https://doi.org/10. 1175/JPO-D-11-0114.1

Cessi, P. (2018). The effect of Northern Hemisphere winds on the meridional overturning circulation and stratification. Journal of Physical Oceanography, 48, 2495–2506. https://doi.org/10.1175/JPO-D-18-0085.1

Cessi, P., & Jones, C. (2017). Warm-route versus cold-route interbasin exchange in the meridional overturning circulation. Journal of Physical Oceanography, 47, 1981–1997.

Cessi, P., & Louazel, S. (2001). Decadal oceanic response to stochastic wind forcing. Journal of Physical Oceanography, 31, 3020-3029.

Cessi, P., & Wolfe, C. L. (2013). Adiabatic eastern boundary currents. Journal of Physical Oceanography, 43(6), 1127-1149.

Cessi, P., Wolfe, C. L., & Ludka, B. C. (2010). Eastern-boundary contribution to the residual and meridional overturning circulations. *Journal of Physical Oceanography*, 40(9), 2075–2090. https://doi.org/10.1175/2010JPO4426.1

Chidichimo, M., Kanzow, T., Cunningham, S., Johns, W., & Marotzke, J. (2010). The contribution of eastern-boundary density variations to the Atlantic meridional overturning circulation at 26.5°N. Ocean Science, 6(2), 475–490.

Cimatoribus, A. A., Drijfhout, S. S., Den Toom, M., & Dijkstra, H. A. (2012). Sensitivity of the Atlantic meridional overturning circulation to South Atlantic freshwater anomalies. *Climate Dynamics*, 39(9-10), 2291–2306.

Cimatoribus, A. A., Drijfhout, S., & Dijkstra, H. A. (2013). Reconciling the north–south density difference scaling for the Meridional Overturning Circulation strength with geostrophy. Ocean Science Discussions, 10, 2461–2479. https://doi.org/10.5194/osd-10-2461-2013 Cimatoribus, A. A., Drijfhout, S. S., & Dijkstra, H. A. (2014). Meridional overturning circulation: Stability and ocean feedbacks in a box

model. Climate Dynamics, 42(1-2), 311-328.

Clément, L., Frajka-Williams, E., Szuts, Z., & Cunningham, S. (2014). Vertical structure of eddies and Rossby waves, and their effect on the Atlantic meridional overturning circulation at 26.5°N. *Journal of Geophysical Research: Oceans*, *119*, 6479–6498. https://doi.org/10. 1002/2014jc010146

Cunningham, S., & coauthors (2019). Pole-to-pole in-situ observing systems for measuring volume, heat and freshwater fluxes of the Atlantic Meridional Overturning Circulation. *Reviews of Geophysics*. submitted. https://doi.org/10:1002/2019RG000661

Cunningham, S. A., Kanzow, T., Rayner, D., Baringer, M. O., Johns, W. E., Marotzke, J., et al. (2007). Temporal variability of the Atlantic meridional overturning circulation at 26.5°N. Science, 317(5840), 935–938.

Czeschel, L., Marshall, D. P., & Johnson, H. L. (2010). Oscillatory sensitivity of Atlantic overturning to high-latitude forcing. *Geophysical Research Letters*, 37, L10601. https://doi.org/10.1029/2010GL043177

Danabasoglu, G., Large, W. G., & Briegleb, B. P. (2010). Climate impacts of parameterized Nordic Sea overflows. Journal of Geophysical Research, 115, C11005. https://doi.org/10.1029/2010JC006243

Danabasoglu, G., McWilliams, J. C., & Gent, P. R. (1994). The role of mesoscale tracer transport in the global ocean circulation. *Science*, 264, 1123–1126.

de Boer, A. M., Gnanadesikan, A., Edwards, N. R., & Watson, A. J. (2010). Meridional density gradients do not control the Atlantic overturning circulation. Journal of Physical Oceanography, 40(2), 368–380. https://doi.org/10.1175/2009JPO4200.1

#### Acknowledgments

H. L. J. and D. P. M. are grateful for funding from the U.K. Natural Environment Research Council under the UK-OSNAP project (NE/K010948/1). P. C. gratefully acknowledges support by the National Science Foundation through OCE-1634128, M. A. S. was supported by the National Science Foundation Grants OCE-1558742 and OCE-1634468. We are also grateful to Eli Tziperman and an anonymous reviewer whose comments helped us to improve the manuscript. The Estimating the Circulation and Climate of the Ocean state estimate (ECCO version 4 release 3) used to produce Figure 2 is available online (https://ecco.jpl.nasa.gov). Please refer to the original papers reviewed here for access to any other data discussed.

- de Jong, F., & de Steur, L. (2016). Strong winter cooling over the Irminger Sea in winter 2014-2015, exceptional deep convection, and the emergence of anomalously low SST. *Geophysical Research Letters* 43, 7106–7113. https://doi.org/10.1002/2016gl069596
- de Lavergne, C., Madec, G., Capet, X., Maze, G., & Roquet, F. (2016). Getting to the bottom of the ocean. Nature Geoscience, 9, 857–858.
  de Lavergne, C., Madec, G., Roquet, F., Holmes, R. M., & McDougall, T. J. (2017). Abyssal ocean overturning shaped by seafloor distribution.
  Nature, 551, 181.
- de Vries, P., & Weber, S. L. (2005). The Atlantic freshwater budget as a diagnostic for the existence of a stable shut down of the meridional overturning circulation. *Geophysical Research Letters*, 32, L09606, https://doi.org/10.1029/2004gl021450
- Dijkstra, H. A. (2007). Characterization of the multiple equilibria regime in a global ocean model. Tellus A: Dynamic Meteorology and Oceanography, 59(5), 695-705.
- Doos, K., Nilsson, J., Nycander, J., Brodeau, L., & Ballarotta, M. (2012). The world ocean thermohaline circulation. Journal of Physical Oceanography, 42(9), 1445–1460. https://doi.org/10.1175/JPO-D-11-0163.1
- Doos, K., & Webb, D. J. (1994). The Deacon Cell and the other meridional cells of the Southern Ocean. *Journal of Physical Oceanography*, 24(2), 429–442.
- Drijfhout, S., Marshall, D., & Dijkstra, H. (2013). Conceptual models of the wind-driven and thermohaline circulation, (2nd Ed). In G. Siedler, S. Griffies, J. Gould, & J. Church (Eds.), *Ocean circulation and climate: A 21st century perspective* (Vol. 103, pp. 257–282). Kidlington, Oxford, UK: Academic Press.
- Drijfhout, S. S., Weber, S. L., & van der Swaluw, E. (2011). The stability of the MOC as diagnosed from model projections for pre-industrial, present and future climates. *Climate Dynamics*, 37(7-8), 1575–1586.
- Duchez, A., Hirschi, J.-M., Cunningham, S., Blaker, A., Bryden, H., de Cuevas, B., et al. (2014). A new index for the Atlantic Meridional Overturning Circulation at 26°N. *Journal of Climate*, *27*(17), 6439–6455.
- Edwards, C. A., & Pedlosky, J. (1998). Dynamics of nonlinear cross-equatorial flow. Part I: Potential vorticity transformation. *Journal of Physical Oceanography*, 28, 2382–2406.
- Emile-Geay, J., Cane, M. A., Naik, N., Seager, R., Clement, A. C., & van Geen, A. (2003). Warren revisited: Atmospheric freshwater fluxes and "Why is no deep water formed in the North Pacific". *Journal of Geophysical Research*, 108(C8), 3178. https://doi.org/10.1029/ 2001jc001058
- Evans, D. G., Toole, J., Forget, G., Zika, J. D., Naveira Garabato, A. C., Nurser, A. J. G., & Yu, L. (2017). Recent wind-driven variability in Atlantic water mass distribution and Meridional Overturning Circulation. *Journal of Physical Oceanography*, 47(3), 633–647. https://doi. org/10.1175/JPO-D-16-0089.1
- Farneti, R., Delworth, T. L., Rosati, A. J., Griffies, S. M., & Zen, F. (2010). The role of mesoscale eddies in the rectification of the Southern Ocean response to climate change. Journal of Physical Oceanography, 40, 1539–1557.
- Ferrari, R., Jansen, M. F., Adkins, J. F., Burke, A., Stewart, A. L., & Thompson, A. F. (2014). Antarctic sea ice control on ocean circulation in present and glacial climates. *Proceedings of the National Academy of Sciences*, 111(24), 8753–8758. https://doi.org/10.1073/pnas. 1323922111
- Ferrari, R., Mashayek, A., McDougall, T. J., Nikurashin, M., & Campin, J.-M. (2016). Turning ocean mixing upside down. Journal of Physical Oceanography, 46(7), 2239–2261.
- Ferrari, R., Nadeau, L.-P., Marshall, D. P., Allison, L. C., & Johnson, H. L. (2017). A model of the ocean overturning circulation with two closed basins and a reentrant channel. *Journal of Physical Oceanography*, 47(12), 2887–2906. https://doi.org/10.1175/JPO-D-16-0223.1
- Ferreira, D., Cessi, P., Coxall, H., de Boer, A., Dijkstra, H., Drijfout, S., et al. (2018). Atlantic-Pacific asymmetry in deep water formation. *Annual Reviews in Earth and Planetary Science*, 46, 327–352.
- Ferreira, D., Marshall, J., & Campin, J.-M. (2010). Localization of deep water formation: Role of atmospheric moisture transport and geometrical constraints on ocean circulation. Journal of Climate, 23(6), 1456–1476. https://doi.org/10.1175/2009JCLI3197.1
- Frajka-Williams, E. (2015). Estimating the Atlantic overturning at 26°N using satellite altimetry and cable measurements. *Geophysical Research Letters*, 42, 3458–3464. https://doi.org/10.1002/2015gl063220
- Fürst, J. J., & Levermann, A. (2012). A minimal model for wind-and mixing-driven overturning: Threshold behavior for both driving mechanisms. Climate Dynamics, 38(1-2), 239–260.
- Gent, P. R. (2018). A commentary on the Atlantic meridional overturning circulation stability in climate models. Ocean Modelling, 122, 57–66. https://doi.org/10.1016/j.ocemod.2017.12.006
- Gent, P. R., Willebrand, J., McDougall, T. J., & McWilliams, J. C. (1995). Parameterizing eddy-induced tracer transports in ocean circulation models. Journal of Physical Oceanography, 25, 463–474.
- Gill, A. (1968). A linear model of the Antarctic Circumpolar Current. Journal of Fluid Mechanics, 32(03), 465-488.
- Gjermundsen, A., & Lacasce, J. H. (2017). Comparing the linear and nonlinear buoyancy-driven circulation. Tellus A: Dynamic Meteorology and Oceanography, 69(1), 1299282.
- Gnanadesikan, A. (1999). A simple predictive model for the structure of the oceanic pycnocline. Science, 283, 2077–2079.
- Goodwin, P. (2012). An isopycnal box model with predictive deep-ocean structure for biogeochemical cycling applications. *Ocean Modelling*, *51*, 19–36. https://doi.org/10.1016/j.ocemod.2012.04.005
- Gordon, A. L. (1986). Interocean exchange of thermocline water. Journal of Geophysical Research, 91(C4), 5037-5046.
- Groeskamp, S., Griffies, S. M., Iudicone, D., Marsh, R., Nurser, A. G., & Zika, J. D. (2019). The water mass transformation framework for ocean physics and biogeochemistry. *Annual Review of Marine Science*, 11(1), 271–305. https://doi.org/10.1146/ annurev-marine-010318-095421
- Groeskamp, S., Zika, J. D., McDougall, T. J., Sloyan, B. M., & Lalibert, F. (2014). The representation of ocean circulation and variability in thermodynamic coordinates. *Journal of Physical Oceanography*, 44(7), 1735–1750. https://doi.org/10.1175/JPO-D-13-0213.1
- Hallberg, R. W., & Gnanadesikan, A. (2001). An exploration of the role of transient eddies in determining the transport of a zonally reentrant current. *Journal of Physical Oceanography*, *31*, 3312–3330.
- Hallberg, R. W., & Gnanadesikan, A. (2006). The role of eddies in determining the structure and response of the wind-driven southern hemisphere overturning: Results from the Modeling Eddies in the Southern Ocean (MESO) project. *Journal of Physical Oceanography*, *36*, 2232–2252.
- Hawkins, E., Smith, R. S., Allison, L. C., Gregory, J. M., Woollings, T. J., Pohlmann, H., & De Cuevas, B. (2011). Bistability of the Atlantic overturning circulation in a global climate model and links to ocean freshwater transport. *Geophysical Research Letters*, 38, L10605. https://doi.org/10.1029/2011gl047208
- Hieronymus, M., Nilsson, J., & Nycander, J. (2014). Water mass transformation in salinity-temperature space. Journal of Physical Oceanography, 44(9), 2547–2568. https://doi.org/10.1175/JPO-D-13-0257.1
- Hirschi, J. J.-M., Blaker, A. T., Sinha, B., Coward, A. C., De Cuevas, B. A., Alderson, S. G., & Madec, G. (2013). Chaotic variability of the meridional overturning circulation on subannual to interannual timescales. *Ocean Science*, 9, 805–823.



Hughes, C. W., Fukumori, I., Griffies, S. M., Huthnance, J. M., Minobe, S., Spence, P., et al. (2019). Sea level and the role of coastal trapped waves in mediating the influence of the open ocean on the coast. *Surveys in Geophysics*. https://doi.org/10.1007/s10712-019-09535-x

Hughes, C. W., Williams, J., Blaker, A., Coward, A., & Stepanov, V. (2018). A window on the deep ocean: The special value of ocean bottom pressure for monitoring the large-scale, deep-ocean circulation. *Progress in Oceanography*, 161, 19–46. https://doi.org/10.1016/j.pocean. 2018.01.011

Huisman, S. E., Den Toom, M., Dijkstra, H. A., & Drijfhout, S. (2010). An indicator of the multiple equilibria regime of the Atlantic meridional overturning circulation. *Journal of Physical Oceanography*, 40(3), 551–567.

- Isachsen, P. E., Mauritzen, C., & Svendsen, H. (2007). Dense water formation in the Nordic Seas diagnosed from sea surface buoyancy fluxes. *Deep Sea Research Part I: Oceanographic Research Papers*, 54(1), 22–41. https://doi.org/10.1016/j.dsr.2006.09.008
- Jia, Y., Coward, A. C., De Cuevas, B. A., Webb, D. J., & Drijfhout, S. S. (2007). A model analysis of the behavior of the Mediterranean water in the North Atlantic. *Journal of Physical Oceanography*, *37*, 764–786.
- Johnson, H. L., & Marshall, D. P. (2002a). A theory for the surface Atlantic response to thermohaline variability. *Journal of Physical Oceanography*, 32(4), 1121–1132.
- Johnson, H. L., & Marshall, D. P. (2002b). Localization of abrupt change in the North Atlantic thermohaline circulation. *Geophysical Research Letters*, 29(6), 1083. https://doi.org/10.1029/2001gl014140
- Johnson, H. L., & Marshall, D. P. (2004). Global teleconnections of meridional overturning circulation anomalies. Journal of Physical Oceanography, 34(7), 1702–1722. https://doi.org/10.1175/1520-0485(2004)034<1702:gtomoc>2.0.co;2
- Johnson, H. L., Marshall, D. P., & Sproson, D. A. (2007). Reconciling theories of a mechanically driven meridional overturning circulation with thermohaline forcing and multiple equilibria. *Climate Dynamics*, 29(7-8), 821–836.
- Jones, C. S., & Cessi, P. (2016). Interbasin transport of the meridional overturning circulation. Journal of Physical Oceanography, 46(4), 1157–1169. https://doi.org/10.1175/JPO-D-15-0197.1
- Kanzow, T., Johnson, H., Marshall, D., Cunningham, S., Hirschi, J.-M., Mujahid, A., et al. (2009). Basinwide integrated volume transports in an eddy-filled ocean. *Journal of Physical Oceanography*, 39(12), 3091–3110.
- Legg, S., Briegleb, B., Chang, Y., Chassignet, E. P., Danabasoglu, G., Ezer, T., et al. (2009). Improving oceanic overflow representation in climate models: The gravity current entrainment climate process team. *Bulletin of the American Meteorological Society*, 90(5), 657–670. https://doi.org/10.1175/2008BAMS2667.1
- Liu, W., & Liu, Z. (2013). A diagnostic indicator of the stability of the Atlantic Meridional Overturning Circulation in CCSM3. Journal of Climate, 26(6), 1926–1938. https://doi.org/10.1175/JCLI-D-11-00681.1
- Liu, W., & Liu, Z. (2014). Assessing the stability of the Atlantic meridional overturning circulation of the past, present, and future. Journal of Meteorological Research, 28(5), 803–819.
- Liu, W., Liu, Z., & Hu, A. (2013). The stability of an evolving Atlantic meridional overturning circulation. *Geophysical Research Letters*, 40, 1562–1568. https://doi.org/10.1002/grl.50365
- Lozier, S. M., Bacon, S., Bower, A. S., Cunningham, S. A., Femke de Jong, M., de Steur, L., et al. (2017). Overturning in the Subpolar North Atlantic Program: A new international ocean observing system. *Bulletin of the American Meteorological Society*, *98*(4), 737–752. https://doi.org/10.1175/BAMS-D-16-0057.1
- Lozier, M. S., Li, F., Bacon, S., Bahr, F., Bower, A. S., Cunningham, S. A., et al. (2019). A sea change in our view of overturning in the subpolar North Atlantic. Science, 363(6426), 516–521. https://doi.org/10.1126/science.aau6592
- Lucarini, V., & Stone, P. H. (2005). Thermohaline circulation stability: A box model study. Part II: Coupled atmosphere-ocean model. Journal of Climate, 18(4), 514–529. https://doi.org/10.1175/JCLI-3279.1
- MacCready, P. (1994). Frictional decay of abyssal boundary currents. Journal of Marine Research, 52(2), 197–217. https://doi.org/10.1357/0022240943077073
- Mackay, N., Wilson, C., Zika, J., & Holliday, N. P. (2018). A regional thermohaline inverse method for estimating circulation and mixing in the Arctic and Subpolar North Atlantic. *Journal of Atmospheric and Oceanic Technology*, 35(12), 2383–2403. https://doi.org/10.1175/ JTECH-D-17-0198.1
- Mak, J., Maddison, J. R., Marshall, D. P., & Munday, D. R. (2018). Implementation of a geometrically informed and energetically constrained mesoscale eddy parameterization in an ocean circulation model. *Journal of Physical Oceanography*, 48, 2363–2382.
- Marotzke, J. (1997). Boundary mixing and the dynamics of three-dimensional thermohaline circulations. *Journal of Physical Oceanography*, 27(8), 1713–1728.
- Marotzke, J., & Scott, J. R. (1999). Convective mixing and the thermohaline circulation. *Journal of Physical Oceanography*, 29, 2962–2970.
  Marsh, R. (2000). Recent variability of the North Atlantic thermohaline circulation inferred from surface heat and freshwater fluxes. *Journal of Climate*, 13(18), 3239–3260. https://doi.org/10.1175/1520-0442(2000)013<3239:RVOTNA>2.0.CO;2
- Marsh, R., Josey, S. A., Nurser, A. J. G., de Cuevas, B. A., & Coward, A. C. (2005). Water mass transformation in the North Atlantic over 1985-2002 simulated in an eddy-permitting model. Ocean Science, 1, 127–144. https://doi.org/10.5194/os-1-127-2005

Marshall, D. (1997). Subduction of water masses in an eddying ocean. Journal of Marine Research, 55, 201-222.

- Marshall, D. P., & Johnson, H. L. (2013). Propagation of meridional circulation anomalies along western and eastern boundaries. Journal of Physical Oceanography, 43(12), 2699–2717.
- Marshall, D. P., & Johnson, H. L. (2017). Relative strength of the Antarctic Circumpolar Current and Atlantic Meridional Overturning Circulation. *Tellus A: Dynamic Meteorology and Oceanography*, 69(1), 1338884. https://doi.org/10.1080/16000870.2017.1338884
- Marshall, D. P., Munday, D. R., Allison, L. C., Hay, R. J., & Johnson, H. L. (2016). Gill's model of the Antarctic Circumpolar Current, revisited: The role of latitudinal variations in wind stress. *Ocean Modelling*, 97, 37–51.
- Marshall, D. P., & Pillar, H. R. (2011). Momentum balance of the wind-driven and meridional overturning circulation. Journal of Physical Oceanography, 41(5), 960–978. https://doi.org/10.1175/2011JPO4528.1
- Marshall, J. C., & Radko, T. (2003). Residual-mean solutions for the Antarctic Circumpolar Current and its associated overturning circulation. *Journal of Physical Oceanography*, *33*, 2341–2354. https://doi.org/10.1175/1520-0485(2003)033<2341:RSFTAC>2.0CO;2 Marshall, J., & Schott, F. (1999). Open-ocean convection: Observations, theory, and models. *Reviews of Geophysics*, *37*, 1–64.
- Marshall, J., & Speer, K. (2012). Closure of the meridional overturning circulation through Southern Ocean upwelling. *Nature Geoscience*, 5, 171. https://doi.org/10.1038/ngeo1391
- Marshall, D. P., & Zanna, L. (2014). A conceptual model of ocean heat uptake under climate change. Journal of Climate, 27(22), 8444–8465. https://doi.org/10.1175/JCLI-D-13-00344.1



McCarthy, G., Smeed, D., Johns, W., Frajka-Williams, E., Moat, B., Rayner, D., et al. (2015). Measuring the Atlantic meridional overturning circulation at 26°N. *Progress in Oceanography*, 130, 91–111.

- McCreary, J. P., Furue, R., Schloesser, F., Burkhardt, T. W., & Nonaka, M. (2016). Dynamics of the Atlantic meridional overturning circulation and Southern Ocean in an ocean model of intermediate complexity. *Progress in Oceanography*, 143, 46–81.
- Mecking, J., Drijfhout, S., Jackson, L., & Andrews, M. (2017). The effect of model bias on Atlantic freshwater transport and implications for AMOC bi-stability. *Tellus A: Dynamic Meteorology and Oceanography*, 69(1), 1299910. https://doi.org/10.1080/16000870.2017.1299910
- Mecking, J. V., Drijfhout, S. S., Jackson, L. C., & Graham, T. (2016). Stable AMOC off state in an eddy-permitting coupled climate model. *Climate Dynamics*, 47(7), 2455–2470. https://doi.org/10.1007/s00382-016-2975-0
- Meinen, C. S., Speich, S., Piola, A. R., Ansorge, I., Campos, E., Kersalé, M., et al. (2018). Meridional Overturning Circulation transport variability at 34.5°S during 2009–2017: Baroclinic and barotropic flows and the dueling influence of the boundaries. *Geophysical Research Letters*, 45, 4180–4188. https://doi.org/10.1029/2018GL077408
- Morrison, A. K., & Hogg, A. M. c. C. (2013). On the relationship between Southern Ocean overturning and ACC transport. Journal of Physical Oceanography, 43(1), 140–148. https://doi.org/10.1175/JPO-D-12-057.1
- Munday, D. R., Johnson, H. L., & Marshall, D. P. (2013). Eddy saturation of equilibrated circumpolar currents. Journal of Physical Oceanography, 43, 507–532.

Munk, W. H. (1966). Abyssal recipes. Deep Sea Research and Oceanographic Abstracts, 13, 707-730.

Munk, W. H., & Palmén, E. (1951). Note on the dynamics of the Antarctic Circumpolar Current. Tellus, 3, 53-55.

- Munk, W., & Wunsch, C. (1998). Abyssal recipes II: Energetics of tidal and wind mixing. Deep Sea Research Part I: Oceanographic Research Papers, 45(12), 1977–2010. https://doi.org/10.1016/S0967-0637(98)00070-3
- Newsom, E. R., & Thompson, A. F. (2018). Reassessing the role of the Indo-Pacific in the ocean's global overturning circulation. Geophysical Research Letters, 45, 12,422–12,431. https://doi.org/10.1029/2018GL080350
- Nieves, D., & Spall, M. (2018). Propagation of North Atlantic Deep Water Anomalies. Journal of Physical Oceanography, 48(8), 1831–1848. https://doi.org/10.1175/JPO-D-18-0068.1
- Nikurashin, M., & Vallis, G. (2012). A theory of the interhemispheric meridional overturning circulation and associated stratification. *Journal of Physical Oceanography*, 42(10), 1652–1667. https://doi.org/10.1175/JPO-D-11-0189.1
- Nilsson, J., Langen, P. L., Ferreira, D., & Marshall, J. (2013). Ocean basin geometry and the salinification of the Atlantic Ocean. Journal of Climate, 26, 6163–6184. https://doi.org/10.1175/JCLI-D-12-00358.1
- Pedlosky, J. (2003). Thermally driven circulations in small ocean basins. Journal of Physical Oceanography, 33, 2333-2340.
- Pedlosky, J., & Spall, M. A. (2005). Boundary intensification of vertical velocity in a β-plane basin. *Journal of Physical Oceanography*, 35(12), 2487–2500.

Pickart, R. S., & Spall, M. A. (2007). Impact of Labrador Sea convection on the North Atlantic meridional overturning circulation. *Journal of Physical Oceanography*, 37, 2207–2227.

Pickart, R. S., Straneo, F., & Moore, G. W. K. (2003). Is Labrador Sea Water formed in the Irminger Basin? Deep-Sea Research Part I, 50, 23–52.

- Pillar, H. R., Heimbach, P., Johnson, H. L., & Marshall, D. P. (2016). Dynamical attribution of recent variability in Atlantic overturning. Journal of Climate, 29(9), 3339–3352.
- Polo, I., Robson, J., Sutton, R., & Balmaseda, M. A. (2014). The importance of wind and buoyancy forcing for the boundary density variations and the geostrophic component of the AMOC at 26°N. *Journal of Physical Oceanography*, 44(9), 2387–2408. https://doi.org/10.1175/ JPO-D-13-0264.1
- Price, J. F., & Baringer, M. O. (1994). Outflows and deep water production by marginal seas. Progress in Oceanography, 33, 161-200.
- Radko, T., & Kamenkovich, I. (2011). Semi-adiabatic model of the deep stratification and meridional overturning. *Journal of Physical Oceanography*, 41(4), 757–780. https://doi.org/10.1175/2010JPO4538.1
- Reid, J. L. (1961). On the temperature, salinity, and density differences between the Atlantic and Pacific oceans in the upper kilometre. Deep Sea Research, 7(4), 265–275. https://doi.org/10.1016/0146-6313(61)90044-2
- Reid, J. L. (1979). On the contribution of the Mediterranean Sea outflow to the Norwegian-Greenland Sea. Deep Sea Research, 26A, 1199–1223.
- Ridgway, K. R., & Dunn, J. R. (2003). Mesoscale structure of the mean East Australian Current System and its relationship with topography. *Progress in Oceanography*, 56, 189–222.
- Robinson, A., & Stommel, H. (1959). The oceanic thermocline and the associated thermohaline circulation 1. Tellus, 11(3), 295-308.
- Samelson, R. M. (2009). A simple dynamical model of the warm-water branch of the mid-depth meridional overturning cell. Journal of
- Physical Oceanography, 39(5), 1216–1230. https://doi.org/10.1175/2008JPO4081.1
  Samelson, R. M. (2011). Time-dependent adjustment in a simple model of the mid-depth meridional overturning cell. Journal of Physical Oceanography, 41(5), 1009–1025. https://doi.org/10.1175/2010JPO4562.1
- Schloesser, F. (2015). Large-scale dynamics of circulations with open-ocean convection. Journal of Physical Oceanography, 45(12), 2933-2951.
- Schloesser, F., Furue, R., McCreary, J. P., & Timmermann, A. (2012). Dynamics of the Atlantic meridional overturning circulation. Part 1: Buoyancy-forced response. *Progress in Oceanography*, 101(1), 33–62. https://doi.org/10.1016/j.pocean.2012.01.002
- Schloesser, F., Furue, R., McCreary, J., & Timmermann, A. (2014). Dynamics of the Atlantic meridional overturning circulation. Part 2: Forcing by winds and buoyancy. *Progress in Oceanography*, *120*, 154–176. https://doi.org/10.1016/j.pocean.2013.08.007
- Schmitt, R. W., Bogden, P. S., & Dorman, C. E. (1989). Evaporation minus precipitation and density fluxes for the North Atlantic. Journal of Physical Oceanography, 19, 1208–1221.
- Schmitz, W. J. (1995). On the interbasin-scale thermohaline circulation. Reviews of Geophysics, 33(2), 151–173.
- Send, U., Lankhorst, M., & Kanzow, T. (2011). Observation of decadal change in the Atlantic meridional overturning circulation using 10 years of continuous transport data. *Geophysical Research Letters*, 38, L24606. https://doi.org/10.1029/2011GL049801
- Send, U., & Marshall, J. C. (1995). Integral effects of deep convection. Journal of Physical Oceanography, 25, 855–872. https://doi.org/ 10.1175/1520-0485(1995)025<0855:IEODC>2.0.CO;2
- Shakespeare, C. J., & Hogg, A. (2012). An analytical model of the response of the meridional overturning circulation to changes in wind and buoyancy forcing. Journal of Physical Oceanography, 42(8), 1270–1287. https://doi.org/10.1175/JPO-D-11-0198.1
- Sijp, W. P. (2012). Characterising meridional overturning bistability using a minimal set of state variables. *Climate Dynamics*, *39*(9-10), 2127–2142.
- Sijp, W. P., Gregory, J. M., Tailleux, R., & Spence, P. (2012). The key role of the western boundary in linking the AMOC strength to the north-south pressure gradient. *Journal of Physical Oceanography*, 42(4), 628–643. https://doi.org/10.1175/JPO-D-11-0113.1

Smeed, D., McCarthy, G., Cunningham, S., Frajka-Williams, E., Rayner, D., Johns, W., et al. (2014). Observed decline of the Atlantic meridional overturning circulation 2004–2012. Ocean Science, 10(1), 29–38.

Spall, M. A. (2003). On the thermohaline circulation in flat bottom marginal seas. Journal of Marine Research, 61, 1–25.

Spall, M. A. (2004). Boundary currents and water mass transformation in marginal seas. Journal of Physical Oceanography, 34, 1197–1213.

Spall, M. A. (2010). Dynamics of downwelling in an eddy-resolving convective basin. Journal of Physical Oceanography, 40, 2341–2347.

Spall, M. A. (2012). Influences of precipitation on water mass transformation and deep convection. *Journal of Physical Oceanography*, 42, 1684–1700.

Spall, M. A., & Pickart, R. S. (2001). Where does dense water sink? A subpolar gyre example. *Journal of Physical Oceanography*, 31(3), 810–826.

Speer, K. G. (1993). Conversion among North Atlantic surface water types. Tellus A, 45, 72-79.

Speer, K., Rintoul, S. R., & Sloyan, B. (2000). The diabatic Deacon Cell. Journal of Physical Oceanography, 30, 3212–3222.

Speer, K., & Tziperman, E. (1992). Rates of water mass formation in the North Atlantic ocean. Journal of Physical Oceanography, 22(1), 93–104. https://doi.org/10.1175/1520-0485(1992)022<0093:ROWMFI>2.0.CO;2

Speich, S., Blanke, B., & Cai, W. (2007). Atlantic meridional overturning circulation and the Southern Hemisphere supergyre. Geophysical Research Letters, 34, L23614. https://doi.org/10.1029/2007GL031583

Srokosz, M., & Bryden, H. (2015). Observing the Atlantic Meridional Overturning Circulation yields a decade of inevitable surprises. Science, 348(6241), 1255575.

Stommel, H. M. (1961). Thermohaline convection with two stable regimes of flow. Tellus, 13, 224-230.

Straneo, F. (2006). On the connection between dense water formation, overturning, and poleward heat transport in a convective basin. *Journal of Physical Oceanography*, *36*, 1822–1840. https://doi.org/10.1175/JPO02932.1

Sumata, H., & Kubokawa, A. (2001). Numerical study of eastern boundary ventilation and its effects on the thermocline structure. Journal of Physical Oceanography, 31(10), 3002–3019.

Talley, L. D. (2013). Closure of the global overturning circulation through the Indian, Pacific, and Southern Oceans: Schematics and transports. *Oceanography*, 26, 80–97. https://doi.org/10.5670/oceanog.2013.07

Tansley, C. E., & Marshall, D. P. (2001). On the dynamics of wind-driven circumpolar currents. Journal of Physical Oceanography, 31, 3258-3273.

te Raa, L. A., & Dijkstra, H. A. (2002). Instability of the thermohaline ocean circulation on interdecadal timescales. *Journal of Physical Oceanography*, *32*, 138–160.

Thomas, M., & Zhai, X. (2013). Eddy-induced variability of the meridional overturning circulation in a model of the North Atlantic. *Geophysical Research Letters*, 40, 2742–2747. https://doi.org/10.1002/grl.50532

Thompson, A. F., Stewart, A. L., & Bischoff, T. (2016). A multi-basin residual-mean model for the global overturning circulation. Journal of Physical Oceanography, 46, 2583–2604.

Toggweiler, J. R., & Samuels, B. (1995). Effect of Drake Passage on the global thermohaline circulation. Deep-Sea Research, 42, 477-500.

Viebahn, J., & Eden, C. (2010). Towards the impact of eddies on the response of the Southern Ocean to climate change. Ocean Modelling, 34, 150–165.

Waldman, R., Bruggemann, N., Bosse, A., Spall, M., Somot, S., & Sevault, F. (2018). Overturning the Mediterranean thermohaline circulation. *Geophysical Research Letters*, 45, 1–9. https://doi.org/10.1029/2018GL078502

Walin, G. (1982). On the relation between sea-surface heat flow and thermal circulation in the ocean. *Tellus*, 34(2), 187–195. https://doi. org/10.1111/j.2153-3490.1982.tb01806.x

Warren, B. A. (1983). Why is no deep water formed in the North Pacific? Journal of Marine Research, 41(2), 327–347.

Weaver, A. J., Sedláček, J., Eby, M., Alexander, K., Crespin, E., Fichefet, T., et al. (2012). Stability of the Atlantic meridional overturning circulation: A model intercomparison. *Geophysical Research Letters*, 39, L20709. https://doi.org/10.1029/2012GL053763

Weber, S. L., Drijfhout, S. S., Abe-Ouchi, A., Crucifix, M., Eby, M., Ganopolski, A., et al. (2007). The modern and glacial overturning circulation in the Atlantic ocean in PMIP coupled model simulations. *Climate of the Past*, *3*(1), 51–64.

Weijer, W., Cheng, W., Drijfhout, S. S., Fedorov, A. V., Hu, A. V., Jackson, L. C., Liu, W., McDonagh, E. L., Mecking, J. V., & Zhang, J. (2019). Stability of the Atlantic Meridional Overturning Circulation: A Review and Synthesis. *Journal of Geophysical Research: Oceans.* https:// doi.org/10.1029/2019JC015083

Welander, P. (1971). The thermocline problem. Philosophical Transactions of the Royal Society A, 270, 415-421.

Williams, R. G., Roussenov, V., Smith, D., & Lozier, M. S. (2014). Decadal evolution of ocean thermal anomalies in the North Atlantic: The effects of Ekman, overturning, and horizontal transport. *Journal of Climate*, 27(2), 698–719. https://doi.org/10.1175/JCLI-D-12-00234.1 Wise, A., Hughes, C. W., & Polton, J. A. (2018). Bathymetric influence on the coastal sea level response to ocean gyres at western boundaries.

Journal of Physical Oceanography, 48(12), 2949–2964. https://doi.org/10.1175/JPO-D-18-0007.1

Wolfe, C. L., & Cessi, P. (2010). What sets the strength of the middepth stratification and overturning circulation in eddying ocean models? Journal of Physical Oceanography, 40(7), 1520–1538. https://doi.org/10.1175/2010jpo4393.1

Wolfe, C. L., & Cessi, P. (2011). The adiabatic pole-to-pole overturning circulation. Journal of Physical Oceanography, 41(9), 1795–1810. https://doi.org/10.1175/2011jpo4570.1

Wolfe, C. L., & Cessi, P. (2014). Salt feedback in the adiabatic overturning circulation. Journal of Physical Oceanography, 44(4), 1175–1194.

Wolfe, C. L., & Cessi, P. (2015). Multiple regimes and low-frequency variability in the quasi-adiabatic overturning circulation. Journal of Physical Oceanography, 45(6), 1690–1708.

Wright, D. G., & Stocker, T. F. (1991). A zonally averaged ocean model for the thermohaline circulation. Part I: Model development and flow dynamics. Journal of Physical Oceanography, 21(12), 1713–1724. https://doi.org/10.1175/1520-0485(1991)021<1713:AZAOMF> 2.0.CO;2

Wright, D. G., Vreugdenhil, C. B., & Hughes, T. M. C. (1995). Vorticity dynamics and zonally averaged ocean circulation models. *Journal of Physical Oceanography*, 25(9), 2141–2154. https://doi.org/10.1175/1520-0485(1995)025<2141:VDAZAO>2.0.CO;2

Wunsch, C. (2008). Mass and volume transport variability in an eddy-filled ocean. Nature Geoscience, 1, 165–168. https://doi.org/10.1038/ngeo126

Wunsch, C., & Ferrari, R. (2004). Vertical mixing, energy, and the general circulation of the oceans. Annual Reviews of Fluid Mechanics, 36, 281–314.

Yang, J. (2015). Local and remote wind stress forcing of the seasonal variability of the Atlantic Meridional Overturning Circulation (AMOC) transport at 26.5°N. Journal of Geophysical Research: Oceans, 120, 2488–2503. https://doi.org/10.1002/2014jc010317

Zanna, L., Heimbach, P., Moore, A. M., & Tziperman, E. (2011). Optimal excitation of interannual Atlantic meridional overturning circulation variability. Journal of Climate, 24(2), 413-427.



Zhai, X., Johnson, H. L., & Marshall, D. P. (2010). Significant sink of ocean-eddy energy near western boundaries. *Nature Geoscience*, *3*, 608–612. https://doi.org/10.1038/ngeo943

Zhai, X., Johnson, H. L., & Marshall, D. P. (2011). A model of Atlantic heat content and sea level change in response to thermohaline forcing. *Journal of Climate*, 24(21), 5619–5632. https://doi.org/10.1175/JCLI-D-10-05007.1

Zhai, X., Johnson, H. L., & Marshall, D. P. (2014). A simple model of the response of the Atlantic to the North Atlantic Oscillation. Journal of Climate, 27(11), 4052–4069.

Zhao, J., & Johns, W. (2014a). Wind-driven seasonal cycle of the Atlantic meridional overturning circulation. *Journal of Physical Oceanography*, 44(6), 1541–1562.

Zhao, J., & Johns, W. (2014b). Wind-forced interannual variability of the Atlantic Meridional Overturning Circulation at 26.5°N. Journal of Geophysical Research: Oceans, 119, 2403–2419. https://doi.org/10.1002/2013jc009407

Zika, J. D., England, M. H., & Sijp, W. P. (2012). The ocean circulation in thermohaline coordinates. *Journal of Physical Oceanography*, 42(5), 708–724. https://doi.org/10.1175/JPO-D-11-0139.1